Author's response to reviews

Title: Increased prevalence and incidence of anemia among adults in transforming rural China: Two cross-sectional surveys

Authors:

Xuecai Wang (dqjk03@126.com)
Zhaofan Wu (13211020069@fudan.edu.cn)
Yue Chen (ychen@uottawa.ca)
Jianfu Zhu (dqzjf@hotmail.com)
Xiaolian Dong (dqjk13@126.com)
Chaowei Fu (cwfu@shmu.edu.cn)
Qingwu Jiang (jiangqw@fudan.edu.cn)

Version: 2 Date: 3 June 2015

Author's response to reviews:

Dear Editor,

Thank you for your response. In revised manuscript, we added line and page numbers and provided a list of abbreviations used.

Thank you for your consideration.

FU Chaowei, B.M., M.M.Sc., Ph.D.
Associate Professor of Epidemiology
138 Yi Xue Yuan Road
Department of Epidemiology
School of Public Health
Fudan University
Shanghai 200032, China
Tel/ Fax: 86 21 54237811
Email: fcw@fudan.edu.cn